Author’s response to reviews

Title: Tracking physical activity using smart phone apps: Assessing the ability of a current app and systematically collecting patient recommendations for future development

Authors:

Jennifer Murphy (j.murphy@imperial.ac.uk)
Trishna Uttamlal (phd17tu@mail.wbs.ac.uk)
Kelly Schmidtkte (K.Schmidtkte@mmu.ac.uk;kelly.ann.schmidtkte@icloud.com)
Ivo Vlaev (Ivo.Vlaev@wbs.ac.uk)
David Taylor (david.taylor@imperial.ac.uk)
Mian Ahmad (mian.ahmad@imperial.ac.uk)
Susan Alsters (s.alsters@imperial.ac.uk)
Sanjay Purkayastha (s.purkayastha@imperial.ac.uk)
Samantha Scholtz (Samantha.Scholtz2@imperial.nhs.uk)
Ramin Ramezani (raminr@ucla.edu)
Ahmed Ahmed (Ahmed.Ahmed@imperial.nhs.uk)
Harvinder Chahal (h.chahal@imperial.ac.uk)
Ara Darzi (a.darzi@imperial.ac.uk)
Alexandra Blakemore (a.blakemore@imperial.ac.uk)

Version: 5 Date: 10 Jan 2020

Author’s response to reviews:

All authors names should now be correct. We are happy to address any further comments as soon as possible.